Author's response to reviews

Title: Resolution of the three dimensional structure of components of the glomerular filtration barrier

Authors:

Kenton P Arkill (k.p.arkill@bham.ac.uk)
Klaus Qvortrup (qvortrup@sund.ku.dk)
Tobias Starborg (tobias.starborg@manchester.ac.uk)
Judith M Mantell (J.Mantell@bristol.ac.uk)
Charles C Michel (c.c.michel@imperial.ac.uk)
Steve J Harper (s.harper@bristol.ac.uk)
Andy HJ Salmon (Andy.Salmon@bristol.ac.uk)
John M Squire (J.M.Squire@bristol.ac.uk)
Dave O Bates (David.Bates@nottingham.ac.uk)
Chris R Neal (chris.neal@bristol.ac.uk)

Version: 2 Date: 14 November 2013

Author's response to reviews:

List of Changes:

Email Addresses for all authors put on front page.
Table put after references (NB caption still with figure captions).
Carlo Knupp added as an author (due to my earlier over-site).

Statement about human samples and ethics has been added in the methods section. NB Consent is not required under HTA legislation as tissues are archived and anonymised and collected from the living.

Additionally a number of grammatical changes have been made.